Title: The prevalence of pain and disability one year post fracture of the distal radius in a UK population: A cross sectional survey

Authors:

   Catherine M Moore (catherine.moore@nottingham.ac.uk)
   Jo Leonardi-Bee (Jo.Leonardi-Bee@nottingham.ac.uk)

Version: 4 Date: 24 July 2008

Author's response to reviews: see over
Dear Sir/Madam,

Re: Publication of the following article: “The prevalence of pain and disability one year post fracture of the distal radius in a UK population: A cross sectional survey”.

I submit the following amendments to be considered for publication in BMC Musculoskeletal Disorders. Please find enclosed the following:

Manuscript: “moore2008manuscriptamendments240708”

I would like to take this opportunity to thank the reviewers for their comments. My responses are as follows:

Reviewer: Heinz G. Endres

No changes required

Reviewer: Jesse Jupiter

Spelling error: ‘medial’ amended to medical

Comment 1: Author response. Comment added in discussion to address this on page 14 (in blue and underlined)

Comment 2: Author response. Comment added in discussion to address this on page 14 (in blue and underlined)

Reviewer: Ruby Grewal

No changes required

The corresponding author is:
Catherine Moore,
Lecturer of Physiotherapy,
Address as above

e-mail Catherine.moore@nottingham.ac.uk
07811 100562

Yours sincerely